Reviewer 1
1. The discussion section has been modified.
2. The History of past illness section has been modified.
3. IMDC score has been added.
4. During this period, combination immunotherapy was recommended due to the increase in tumour size. However, the patient rejected it for economic reasons.

Reviewer 2
1. Relevant articles have been cited.

In addition, this paper is polished again